---
title: "Overview"
description: "How to use Infisical to inject secrets and configs into various 3-rd party services and frameworks."
---

Integrations allow environment variables to be synced from Infisical into your local development workflow, CI/CD pipelines, and production infrastructure.

Missing an integration? [Throw in a request](https://github.com/Infisical/infisical/issues).

| Integration                                                                           | Type                   | Status                             |
| ------------------------------------------------------------------------------------- | ---------------------- | ---------------------------------- |
| [Docker](/integrations/platforms/docker)                                              | Platform               | Available                          |
| [Docker-Compose](/integrations/platforms/docker-compose)                              | Platform               | Available                          |
| [Kubernetes](/integrations/platforms/kubernetes)                                      | Platform               | Available                          |
| [Terraform](https://registry.terraform.io/providers/Infisical/infisical/latest/docs)  | Infrastructure as code | Available                          |
| [PM2](/integrations/platforms/pm2)                                                    | Platform               | Available                          |
| [Heroku](/integrations/cloud/heroku)                                                  | Cloud                  | Available                          |
| [Vercel](/integrations/cloud/vercel)                                                  | Cloud                  | Available                          |
| [Netlify](/integrations/cloud/netlify)                                                | Cloud                  | Available                          |
| [Render](/integrations/cloud/render)                                                  | Cloud                  | Available                          |
| [Laravel Forge](/integrations/cloud/laravel-forge)                                    | Cloud                  | Available                          |
| [Railway](/integrations/cloud/railway)                                                | Cloud                  | Available                          |
| [Terraform Cloud](/integrations/cloud/terraform-cloud)                                | Cloud                  | Available                          |
| [TeamCity](/integrations/cloud/teamcity)                                              | Cloud                  | Available                          |
| [Fly.io](/integrations/cloud/flyio)                                                   | Cloud                  | Available                          |
| [Supabase](/integrations/cloud/supabase)                                              | Cloud                  | Available                          |
| [Northflank](/integrations/cloud/northflank)                                          | Cloud                  | Available                          |
| [Cloudflare Pages](/integrations/cloud/cloudflare-pages)                              | Cloud                  | Available                          |
| [Cloudflare Workers](/integrations/cloud/cloudflare-workers)                          | Cloud                  | Available                          |
| [Checkly](/integrations/cloud/checkly)                                                | Cloud                  | Available                          |
| [Qovery](/integrations/cloud/qovery)                                                  | Cloud                  | Available                          |
| [HashiCorp Vault](/integrations/cloud/hashicorp-vault)                                | Cloud                  | Available                          |
| [AWS Parameter Store](/integrations/cloud/aws-parameter-store)                        | Cloud                  | Available                          |
| [AWS Secrets Manager](/integrations/cloud/aws-secret-manager)                         | Cloud                  | Available                          |
| [Azure Key Vault](/integrations/cloud/azure-key-vault)                                | Cloud                  | Available                          |
| [GCP Secret Manager](/integrations/cloud/gcp-secret-manager)                          | Cloud                  | Available                          |
| [Windmill](/integrations/cloud/windmill)                                              | Cloud                  | Available                          |
| [Bitbucket](/integrations/cicd/bitbucket)                                             | CI/CD                  | Available                          |
| [Codefresh](/integrations/cicd/codefresh)                                             | CI/CD                  | Available                          |
| [GitHub Actions](/integrations/cicd/githubactions)                                    | CI/CD                  | Available                          |
| [GitLab](/integrations/cicd/gitlab)                                                   | CI/CD                  | Available                          |
| [CircleCI](/integrations/cicd/circleci)                                               | CI/CD                  | Available                          |
| [Travis CI](/integrations/cicd/travisci)                                              | CI/CD                  | Available                          |
| [Rundeck](/integrations/cicd/rundeck)                                                 | CI/CD                  | Available                          |
| [Octopus Deploy](/integrations/cicd/octopus-deploy)                                   | CI/CD                  | Available                          |
| [React](/integrations/frameworks/react)                                               | Framework              | Available                          |
| [Vue](/integrations/frameworks/vue)                                                   | Framework              | Available                          |
| [Express](/integrations/frameworks/express)                                           | Framework              | Available                          |
| [Next.js](/integrations/frameworks/nextjs)                                            | Framework              | Available                          |
| [NestJS](/integrations/frameworks/nestjs)                                             | Framework              | Available                          |
| [SvelteKit](/integrations/frameworks/sveltekit)                                       | Framework              | Available                          |
| [Nuxt](/integrations/frameworks/nuxt)                                                 | Framework              | Available                          |
| [Gatsby](/integrations/frameworks/gatsby)                                             | Framework              | Available                          |
| [Remix](/integrations/frameworks/remix)                                               | Framework              | Available                          |
| [Vite](/integrations/frameworks/vite)                                                 | Framework              | Available                          |
| [Fiber](/integrations/frameworks/fiber)                                               | Framework              | Available                          |
| [Django](/integrations/frameworks/django)                                             | Framework              | Available                          |
| [Flask](/integrations/frameworks/flask)                                               | Framework              | Available                          |
| [Laravel](/integrations/frameworks/laravel)                                           | Framework              | Available                          |
| [Ruby on Rails](/integrations/frameworks/rails)                                       | Framework              | Available                          |
| Jenkins                                                                               | CI/CD                  | Available                          |
